Author's response to reviews

Title: Rural residence is not a risk factor for frequent mental distress: a behavioral risk factor surveillance survey

Authors:

James E Rohrer Ph.D. (james.rohrer@ttuhsc.edu)
Tyrone F Borders Ph.D (tborders@unt.hsc.edu)
Jimmy Blanton MPA (Jimmy.Blanton@dshs.state.tx.us)

Version: 3 Date: 9 May 2005

Author's response to reviews:

Thank you for accepting our manuscript. You asked for some minor editorial changes. here is what we did:

1. Fix bouncing email addresses. Response: I tested them and they worked for me.
2. Remove authors qualifications. Response: the cover page does not show any author's qualifications.
5. Remove tabs from tables and add black borders. Response: I highlighted each table and removed all tabs. However, there must not have been any tabs becasue the tables were unchanged. Borders appear to be black. Please advise if the tables require changes.

thanks.